Author's response to reviews

Title: An unusual anatomical variation of superior turbinate: case report

Authors:

Turhan San (turhansan@gmail.com)
Emre Gürkan (dremre_grkn@yahoo.com)
Barış Erdoğan (bariserdogan1981@yahoo.com)

Version: 6 Date: 10 February 2014

Author's response to reviews:

Title: An unusual anatomical variation of superior turbinate: case report
Authors:

Turhan San (turhansan@gmail.com)
Emre Gürkan (dremre_grkn@yahoo.com)
Barış Erdoğan (bariserdogan1981@yahoo.com)

Version: 2 Date: 9 February 2014

Author's response to reviews

1. The title was changed, added the word ‘anatomical’. (An unusual anatomical variation of superior turbinate: case report)
2. The view of the superior turbinate in the axial plane added as figure 2.
3. Several sentences regarding the causes and pathophysiology of rhinogenic headache added in the discussion section.
4. I’ve upload the figures as separate files via the online submission system.
5. I’ve placed the figure title and legend section after the references list.
6. I’ve added an acknowledgement section at the end of the manuscript before the reference list.